Therapeutic Drug Review on Romosozumab: The First Sclerostin Inhibitor

VIJAY KUMAR JAIN¹, KARTHIKEYAN P IYENGAR², ARVIND NUNE³, GAURAV KUMAR UPADHYAYA⁴

(CC) BY-NC-ND

ABSTRACT

Orthopaedics Section

Osteoporosis is a progressive skeletal disorder which is characterised by low bone mass, normal mineralisation and abnormal bone microarchitecture. This disruption of bone microarchitecture causes subsequent increase in bone fragility and raises risk of fractures. Osteoporosis is a growing public health problem and affects approximately over 200 million people worldwide. In the United Kingdom, it is estimated that around 3 million people have osteoporosis. Since the evolution of drug therapy for osteoporosis in the 1940's with oestrogen therapy several approaches to developing novel therapeutics for osteoporosis in animal studies and clinical observations (e.g., oestrogen, calcitonin, and teriparatide) or opportunistic repurposing of existing compounds (e.g., bisphosphonates) to one driven by advances in fundamental bone biology (e.g., denosumab). The advent of biologic agents has provided a more specific and targeted approach to the treatment of osteoporosis. Sclerostin is a glycoprotein secreted by osteocytes and regulates bone metabolism by inhibiting activation of osteoblast function and bone formation. By inhibiting sclerostin, targeted therapeutic pharmacological agents are being developed to address severe osteoporosis and patients who do not respond well to primary line of medical management of osteoporosis. Romosozumab is humanised as a monoclonal antibody designed to target sclerostin. This review assesses the mechanism and current role of romosozumab in osteoporosis treatment.

Keywords: Antibodies, Bone density, Bone density conservation agents, Monoclonal, Osteocytes, Osteoporosis, Therapy

INTRODUCTION

Osteoporosis is defined as having a Bone Mineral Density (BMD) that is 2.5 standard deviations or more below the average value for young healthy adults (usually referred to as a 'T-score' of -2.5 or lower) [1]. Pharmacological drugs available for the treatment of osteoporosis can be divided into two categories: antiresorptive (or anticatabolic) and anabolic (bone forming) agents. Medical management of osteoporosis usually follows the recommended guidelines [2]. Bone-forming agents are used for shorter durations of treatment, often in patients at extremely high risk of fracture, whereas antiresorptive agents are used as long-term treatments and sometimes after bone-forming agents [3]. Typical medications are initiated depending on intervention thresholds defined using age, T-score and several risk factors.

Clinical prediction tools such as Fracture Risk Assessment Tool (FRAX), osteoporosis self-assessment tool and the Khon Kaen osteoporosis study score for assessing people at risk for osteoporosis are used in planning management [3-5]. The challenge appears when patients do not respond to traditional line of management or have severe osteoporosis at risk of fragility fractures and unable to tolerate conventional agents. The advent of biologic agents has provided a more specific and targeted approach to the treatment of osteoporosis.

ROMOSOZUMAB

Mechanism of Action

Sclerostin is a glycoprotein secreted by osteocytes has been acknowledged found to play a critical role in bone metabolism by inhibiting activation of osteoblast function and bone formation. It inhibits Wnt signalling in osteoblast lineage cells, leading to decreased bone formation by osteoblasts and increased bone resorption by osteoclasts [6].

The Wnt signalling pathways are a group of signal transduction pathways which begin with proteins that pass signals into a cell through cell surface receptors and affect cell function. Wnt binds to LDL (Low Density Lipoprotein)-Receptor-related Protein (LRP) LRP5 and LRP6 co-receptors and Frizzled family receptor leading to activation of the Wnt signalling pathway in osteoblasts and bone formation. By inhibiting sclerostin, Romosozumab stimulates osteoblastic activity and increases bone formation [7].

Romosozumab is a bone-forming agent and a sclerostin antibody that exerts a dual effect on bone, increasing bone formation by osteoblastic activity and decreasing bone resorption resulting in insignificant improvement in BMD as compared to other available therapies. This dual action of the treatment with romosozumab leads to more pronounced increases in BMD than other treatment modalities and reduces the risk of vertebral and clinical fractures by 73% and 36% compared to placebo after 12 months and the sequential treatment regime [8]. Romosozumab increases bone formation by promoting osteoblast differentiation and activity through conversion of bone lining cells to osteoblasts, osteoblastic differentiation of osteoprogenitor cells, and increased bone matrix production by mature osteoblasts [9]. Romosozumab also decreases bone resorption by altering expression of osteoclast regulators including decreasing osteoclast activators RANK (Receptor Activator of Nuclear factor Kappa-B Ligand) and Colony Stimulating Factor-1 (CSF-1) and increasing osteoclast inhibitors OPG (Osteoprotegerin) and WISP (Wnt-1-Induced Secreted Protein-1) [10]. Romosozumab thus is involved in increasing bone formation and decreasing bone resorption.

Romosozumab has been granted marketing authorization by major regulatory authorities in the world including Japan, approved by the Food and Drug Administration (FDA) the USA, Korea, Canada, Australia, and the European Union [11]. It has recently been approved by the National Institute for Care and Health Excellence (United Kingdom) in patients with severe osteoporosis postmenopausal women at increased risk of fractures [12].

In India, recently the Central Drugs Standard Control Organisation, Expert Committee have recommended for grant of permission for import and marketing of romosozumab in the treatment of postmenopausal women with osteoporosis who are at a high risk for fracture. This is defined as a patient with a history of osteoporotic fracture, or having multiple risk factors for fracture; or patients who have failed or are intolerant to other available osteoporosis therapy [13].

Dosages

Romosozumab injection is available in solution form for subcutaneous injection supplied in a single use prefilled syringe which delivers 1.17 mL of a solution containing 105 mg of romosozumab. For an adult, 210 mg once a month for 12 months, supplement with calcium and vitamin D, it is to be administered as two consecutive 105 mg injections at different injection sites into the thigh, abdomen or upper arm [14]. This is to ensure adequate levels of calcium and vitamin D whilst undergoing treatment with romosozumab. It is advisable to correct hypocalcaemia before therapy is initiated. Monitoring for signs and symptoms of hypocalcaemia during therapy is essential [14].

Handling and Storage

It is advised to store romosozumab in a refrigerator between 2-8°C and protect it from light [12].

Therapeutic Uses

Romosozumab is predominantly recommended in postmenopausal women with severe osteoporosis and at increased risk of fractures [12]. Recent guidelines from the American Association of Clinical Endocrinologists and American College of Endocrinology have given a route map of when romosozumab can be used [15]. Romosozumab, which has anabolic effect with monoclonal antibody properties has a significant role in high-risk patients with osteoporosis [16]. Romosozumab has found to be superior to alendronic acid and placebo combination in reducing the incidence of vertebral fractures [17]. The beneficial effect of Romosozumab declines after 12 monthly doses, so its duration of use should be limited to 12 months. For further treatment, antiresorptive agents should be considered. A zoledronate follow-on regimen can maintain robust BMD gains achieved with romosozumab treatment [18]. Romosozumab for 12 months followed by alendronate reduced the risk of vertebral, non vertebral, and hip fractures by 48%, 20% and 38%, respectively compared to alendronate after 2-3 years [8]. Romosozumab also has a crucial role in patients with severe, refractory osteoporosis who have already tried Teriparatide for 18 months. This drug has an advantage over Teriparatide in that Romosozumab can be used in postmenopausal osteoporosis patients who had previous radiation exposure [19].

Side-effects [14]

- Common or quite common: Arthralgia; headache; hypersensitivity; increased risk of infection; muscle spasms; neck pain; skin reactions.
- **Uncommon:** Cataract; hypocalcaemia; myocardial infarction; stroke.
- Rare or exceedingly rare: Angioedema.
- Frequency not known: Atypical femur fracture; cardiovascular event; osteonecrosis of jaw.

Precautions

 Major Adverse Cardiac Events (MACE): Romosozumab use was associated with an increased risk of MACE, a composite of total mortality, myocardial infarction, cerebrovascular disease, and heart failure-related hospitalisation, when compared to alendronate treatment [20]. When compared to alendronate medication, rosozumab use was associated with an elevated risk of MACE, an aggregate of mortality risk, myocardial infarction, stroke, and heart failure-related hospitalisation [20]. Those who suffered a heart attack or cerebrovascular event 12 months before should not be started on romosozumab. If a patient has a heart attack or a cerebrovascular disease while on romosozumab, the medication should be stopped. In a patient with history of myocardial infarction and ischaemic stroke, romosozumab should be used judiciously. The cause of the cardiovascular effect is due to its role in arterial calcification due to increased Sclerostin expression in smooth muscle tissue in areas of vascular calcification [21].

- Osteonecrosis of the jaw: Osteonecrosis of the Jaw (ONJ) has been reported in patients receiving romosozumab [12]. A routine oral and dental examination is required prior to the initiation of romosozumab treatment.
- 3. Atypical Femoral Fractures (AFF): Atypical femoral fractures were described in patients receiving romosozumab just like they have in those taking other bisphosphonates [22]. It is important to rule out the possibility of an incomplete fracture due to an atypical fracture in any patient who complains of pain in their thighs. When a patient develops an atypical femoral fracture, they must be screened for signs and symptoms of femoral fracture in the contralateral leg. Patients should report new or unusual discomfort in the groin, hip, and thigh during treatment [23]. Consideration should be given to stopping romosozumab treatment depending on benefit-risk analysis [24].

Contraindications to Romosozumab Use

- **Hypocalcaemia:** Romosozumab is contraindicated in patients with hypocalcaemia. It should be corrected prior to initiating treatment with romosozumab. Patients should be monitored for signs and symptoms of hypocalcaemia. Patients should be adequately supplemented with calcium and vitamin D while on romosozumab [25].
- History of myocardial infarction or stroke: It is advisable to discontinue romosozumab if myocardial infarction or stroke occurs during treatment [26].
- Systemic hypersensitivity: Any patient who has had an adverse reaction to romosuzumab in the past, whether it was to romosozumab or any of its formulation, should not be treated with it. If the patient experiences an anaphylactic or allergic reaction while taking romosuzumab, the patient should stop taking the drug immediately [27].

Monitoring of Patient Parameters [25]

- 1. It is advisable to monitor serum calcium concentration in patients with severe renal impairment, or in those receiving dialysis- increased risk of hypocalcaemia.
- It is advisable to correct hypocalcaemia before therapy is initiated. It is also prudent to monitor for signs and symptoms of hypocalcaemia during therapy.

CONCLUSION(S)

Romosozumab with its unique mechanism of action provides a significant alternative for the treatment of severe osteoporosis. Romosozumab reduces the risk of vertebral and clinical fractures in women with postmenopausal osteoporosis, with a favourable balance of benefits and risks. Romosozumab is a promising emerging anabolic agent based on favourable results from several phase III trials in postmenopausal women with osteoporosis, and a single trial in men with osteoporosis. Romosozumab may thus expand the options for treating osteoporotic patients at high risk of fracture.

REFERENCES

- [1] Sözen T, Özışık L, Başaran N. An overview and management of osteoporosis. Eur J Rheumatol. 2017;4(1):46-56.
- [2] National Institute for Care and Excellence. Osteoporosis: Assessing the risk of fragility fracture National Institute for Care and Excellence. 2017. Available from: https://www.nice.org.uk/guidance/cg146.

- [3] World Health Organisation. WHO SCIENTIFIC GROUP ON THE ASSESSMENT OF OSTEOPOROSIS AT PRIMARY HEALTH CARE LEVEL. World Health Organisation. 2004. [Available from: https://www.who.int/chp/topics/Osteoporosis.pdf.
- [4] Skedros JG, Sybrowsky CL, Stoddard GJ. The osteoporosis self-assessment screening tool: A useful tool for the orthopaedic surgeon. J Bone Joint Surg Am. 2007;89(4):765-72.
- [5] Pongchaiyakul C, Wanothayaroj E. Performance of the Khon Kaen Osteoporosis Study (KKOS) score for identifying osteoporosis in men. J Med Assoc Thai. 2007;90(8):1518-23.
- [6] Karner CM, Long F. Wnt signaling and cellular metabolism in osteoblasts. Cell Mol Life Sci. 2017;74(9):1649-57.
- [7] Hamaya E. Pharmacological characteristics and clinical study results of romosozumab (EVENITY(®); genetical recombination), a drug with novel mechanism of action to treat osteoporosis at high risk of fracture. Nihon Yakurigaku Zasshi. 2020;155(4):258-67.
- [8] Langdahl B. Treatment of postmenopausal osteoporosis with bone-forming and antiresorptive treatments: Combined and sequential approaches. Bone. 2020;139:115516.
- [9] Sølling ASK, Harsløf T, Langdahl B. The clinical potential of romosozumab for the prevention of fractures in postmenopausal women with osteoporosis. Ther Adv Musculoskelet Dis. 2018;10(5-6):105-15.
- [10] Glantschnig H, Fisher JE, Wesolowski G, Rodan GA, Reszka AA. M-CSF, TNFalpha and RANK ligand promote osteoclast survival by signaling through mTOR/S6 kinase. Cell Death Differ. 2003;10(10):1165-77.
- [11] European Commission Approves EVENITY® (romosozumab) For The Treatment of Severe Osteoporosis In Postmenopausal Women At High Risk of Fracture. 2019.
- [12] National institute for Care and Excellence. Romosozumab [Available from: https://bnf.nice.org.uk/drug/romosozumab.html.
- [13] CDSC Organisation. Romosozumab 2020 [Available from: https://cdsco.gov.in/ opencms/opencms/system/modules/CDSCO.WEB/elements/com.
- [14] Food and Drug Administration. Romosozumab Medication Guide 2019 [Available from: https://www.accessdata.fda.gov/drugsatfda_docs/label/2019/ 761062s000lbl.pdf.
- [15] Camacho PM, Petak SM, Binkley N, Diab DL, Eldeiry LS, Farooki A, et al. American Association of Clinical Endocrinologists/American College of Endocrinology Clinical Practice Guidelines for the diagnosis and treatment of postmenopausal osteoporosis-2020 update. Endocr Pract. 2020;26(Suppl 1):01-46.

- [16] Liu Y, Cao Y, Zhang S, Zhang W, Zhang B, Tang Q, et al. Romosozumab treatment in postmenopausal women with osteoporosis: A meta-analysis of randomized controlled trials. Climacteric. 2018;21(2):189-95.
- [17] Fukumoto S, Matsumoto T. Recent advances in the management of osteoporosis. F1000Res. 2017;6:625.
- [18] McClung MR, Bolognese MA, Brown JP, Reginster JY, Langdahl BL, Maddox J, et al. A single dose of zoledronate preserves bone mineral density for up to 2 years after a second course of romosozumab. Osteoporos Int. 2020;31(11):2231-41.
- [19] Genant HK, Engelke K, Bolognese MA, Mautalen C, Brown JP, Recknor C, et al. Effects of Romosozumab Compared With Teriparatide on Bone Density and Mass at the Spine and Hip in Postmenopausal Women With Low Bone Mass. J Bone Miner Res. 2017;32(1):181-87.
- [20] Choi BG, Rha SW, Yoon SG, Choi CU, Lee MW, Kim SW. Association of major adverse cardiac events up to 5 years in patients with chest pain without significant coronary artery disease in the Korean population. J Am Heart Assoc. 2019;8(12):e010541.
- [21] Saag KG, Petersen J, Brandi ML, Karaplis AC, Lorentzon M, Thomas T, et al. Romosozumab or Alendronate for fracture prevention in women with osteoporosis. N Engl J Med. 2017;377(15):1417-27.
- [22] Cosman F, Crittenden DB, Adachi JD, Binkley N, Czerwinski E, Ferrari S, et al. Romosozumab Treatment in Postmenopausal Women with Osteoporosis. N Engl J Med. 2016;375(16):1532-43.
- [23] Donnelly E, Saleh A, Unnanuntana A, Lane JM. Atypical femoral fractures: Epidemiology, etiology, and patient management. Curr Opin Support Palliat Care. 2012;6(3):348-54.
- [24] Van de Laarschot DM, McKenna MJ, Abrahamsen B, Langdahl B, Cohen-Solal M, Guañabens N, et al. Medical Management of Patients After Atypical Femur Fractures: A Systematic Review and Recommendations From the European Calcified Tissue Society. J Clin Endocrinol Metab. 2020;105(5):1682-99.
- [25] Haas AV, LeBoff MS. Osteoanabolic agents for osteoporosis. Journal of the Endocrine Society. 2018;2(8):922-32.
- [26] National Institute for Care and Health Excellence (NICE). Romosozumab. Available from: https://bnf.nice.org.uk/drug/romosozumab.html.
- [27] Blakely KK, Johnson C. New osteoporosis treatment means new bone formation. Nurs Womens Health. 2020;24(1):52-57.

PARTICULARS OF CONTRIBUTORS:

- 1. Professor, Department of Orthopaedics, Atal Bihari Vajpayee Institute of Medical Sciences, Dr. Ram Manohar Lohia Hospital, Delhi, India.
- 2. Trauma and Orthopaedic Surgeon, Department of Orthopaedics, Southport and Ormskirk NHS Trust, Southport, Lancashire, United Kingdom.
- 3. Consultant, Department of Rheumatology, Southport and Ormskirk NHS Trust, Southport, Lancashire, United Kingdom.
- 4. Associate Professor, Department of Orthopaedics, All India Institute of Medical Sciences, Raebareli, Uttar Pradesh, India.

NAME, ADDRESS, E-MAIL ID OF THE CORRESPONDING AUTHOR:

Gaurav Kumar Upadhyaya, Associate Professor, Department of Orthopaedics, All India Institute of Medical Sciences, Raebareli-229405, Uttar Pradesh, India.

E-mail: drgkupadhyaya@yahoo.co.in

AUTHOR DECLARATION:

- Financial or Other Competing Interests: None
- Was Ethics Committee Approval obtained for this study? NA
- Was informed consent obtained from the subjects involved in the study? NA
- For any images presented appropriate consent has been obtained from the subjects. NA

PLAGIARISM CHECKING METHODS: [Jain H et al.]

- Plagiarism X-checker: Apr 29, 2021
- Manual Googling: Aug 09, 2021
- iThenticate Software: Feb 05, 2022 (16%)

Date of Submission: Apr 26, 2021 Date of Peer Review: Aug 10, 2021 Date of Acceptance: Nov 30, 2021 Date of Publishing: Apr 01, 2022

ETYMOLOGY: Author Origin